Author's response to reviews

Title: The effect of Lipoxin A4 on the interaction between macrophage and osteoblast: possible role in the treatment of aseptic loosening

Authors:

Gang Li (ligang30003000@yahoo.com.cn)
Ping Wu (ping.wu@hsc.utah.edu)
Yao Xu (xuyaoinxiaoyan@tom.com)
Yan Yu (yuyan831214@sina.com)
Li Sun (litchisun@yahoo.com.cn)
Liang Zhu (Owen19840905@yahoo.com.cn)
Duyun Ye (yedy@mails.tjmu.edu.cn)
Jianchao Wang (wangjcwh@yahoo.com.cn)

Version: 3 Date: 19 May 2009

Author's response to reviews: see over
Dear editor,

We are glad to know that our manuscript “1748174306252996” has been accepted.

Following is the point-to-point description of the changes we made to the most recent version of the article which was accessed at the link you provided.

**Affiliations:**
According to your request, we deleted the name of state in No.2 affiliation. **Importantly,** the eighth author (also the last author) of the last version changed his institution. So, his name was deleted. We also deleted his original institution which was the fifth in the last version and his email address.

**Authors’ contributions**
According to your request, names of all the authors had been changed to initials.

**Reference**
We provided the full and accurate details for reference 3, 24 and 34.

**Please present all text in black**
It has been done.

**The manuscript refers to a removed fig 7. Please amend this.**
We are not quite sure about the meaning of this problem. In revised version, we did not upload fig 7. And, in the revised manuscript, fig 7 did not appear. If still anything wrong with fig 7, please let us know. Thank you!

We did not correct anything else.

Sincerely,

Ping Wu, Ph.D.
Health Sciences Center and VA Medical Center,
University of Utah,
Salt Lake City, Utah, 84132, USA,
Telephone: 001-801-582-1565, extension,2299
Email: ping.wu@hsc.utah.edu